Author’s response to reviews

Title: Translating and validating a training needs assessment tool into Greek

Authors:

Adelais Markaki (adamarkaki@yahoo.com)
Nikos Antonakis (antonakisnikos@yahoo.gr)
Carolyn M Hicks (c.m.hicks@bham.ac.uk)
Christos Lionis (lionis@galinos.med.uoc.gr)

Version: 6 Date: 1 May 2007

Author’s response to reviews:

Upon proofreading the accepted manuscript, we have corrected some minor typographical errors in the Acknowledgement section.